

Missouri University of Science and Technology Scholars' Mine

Civil, Architectural and Environmental Engineering Faculty Research & Creative Works Civil, Architectural and Environmental Engineering

01 Dec 2017

Amplification of Earthquake Ground Motions in Washington, DC, and Implications for Hazard Assessments in Central and Eastern North America

Thomas L. Pratt

J. Wright Horton Jr.

Jessica Munoz

Susan E. Hough

et. al. For a complete list of authors, see https://scholarsmine.mst.edu/civarc_enveng_facwork/1435

Follow this and additional works at: https://scholarsmine.mst.edu/civarc_enveng_facwork

Part of the Geotechnical Engineering Commons

Recommended Citation

T. L. Pratt et al., "Amplification of Earthquake Ground Motions in Washington, DC, and Implications for Hazard Assessments in Central and Eastern North America," *Geophysical Research Letters*, vol. 44, no. 24, pp. 12,150-12,160, American Geophysical Union (AGU), Dec 2017. The definitive version is available at https://doi.org/10.1002/2017GL075517

This Article - Journal is brought to you for free and open access by Scholars' Mine. It has been accepted for inclusion in Civil, Architectural and Environmental Engineering Faculty Research & Creative Works by an authorized administrator of Scholars' Mine. This work is protected by U. S. Copyright Law. Unauthorized use including reproduction for redistribution requires the permission of the copyright holder. For more information, please contact scholarsmine@mst.edu.

@AGUPUBLICATIONS



Geophysical Research Letters

RESEARCH LETTER

10.1002/2017GL075517

Key Points:

- Recordings from a temporary seismometer array document amplifications of earthquake ground motions in Washington, DC
- Damage in Washington, DC, during the 2011 M_w 5.8 Mineral, VA, earthquake was likely increased by these locally amplified ground motions
- Amplification of ground motions in other central and eastern North American cities could raise ground motions from moderate earthquakes to damaging levels

Supporting Information:

Supporting Information S1

Correspondence to:

T. L. Pratt, tpratt@usgs.gov

Citation:

Pratt, T. L., Horton, J. W., Jr., Muñoz, J., Hough, S. E., Chapman, M. C., & Olgun, C. G. (2017). Amplification of earthquake ground motions in Washington, DC, and implications for hazard assessments in central and eastern North America. *Geophysical Research Letters*, 44, 12,150–12,160. https://doi.org/10.1002/ 2017GL075517

Received 31 AUG 2017 Accepted 22 NOV 2017 Accepted article online 30 NOV 2017 Published online 23 DEC 2017

The copyright line on this article was changed on 13 MAR 2018 after original online publication.

©2017. American Geophysical Union. All Rights Reserved.

Amplification of Earthquake Ground Motions in Washington, DC, and Implications for Hazard Assessments in Central and Eastern North America

Thomas L. Pratt¹, J. Wright Horton Jr.¹, Jessica Muñoz¹, Susan E. Hough², Martin C. Chapman³, and C. Guney Olgun³

¹U.S. Geological Survey, Reston, VA, USA, ²U.S. Geological Survey, Pasadena, CA, USA, ³Virginia Tech, Blacksburg, VA, USA

Abstract The extent of damage in Washington, DC, from the 2011 M_w 5.8 Mineral, VA, earthquake was surprising for an epicenter 130 km away; U.S. Geological Survey "Did-You-Feel-It" reports suggest that Atlantic Coastal Plain and other unconsolidated sediments amplified ground motions in the city. We measure this amplification relative to bedrock sites using earthquake signals recorded on a temporary seismometer array. The spectral ratios show strong amplification in the 0.7 to 4 Hz frequency range for sites on sediments. This range overlaps with resonant frequencies of buildings in the city as inferred from their heights, suggesting amplification at frequencies to which many buildings are vulnerable to damage. Our results emphasize that local amplification can raise moderate ground motions to damaging levels in stable continental regions, where low attenuation extends shaking levels over wide areas and unconsolidated deposits on crystalline metamorphic or igneous bedrock can result in strong contrasts in near-surface material properties.

Plain Language Summary Shaking during earthquakes in geologically older continental regions like central and eastern North America extends for much greater distances than in younger regions like much of western North America. We show that amplification of ground motions by shallow layers of sediment beneath Washington, DC, likely was responsible for amplifying moderate ground motions during the M_w 5.8 Virginia earthquake in 2011 to damaging levels, despite the earthquake being relatively distant and only moderate in size. This study thus emphasizes the importance of local amplification effects in causing damage to cities in stable continental regions, where there can be strong contrasts in material properties between shallow sediments and underlying igneous or metamorphic bedrock.

1. Introduction

Washington, DC, had surprisingly high intensities of ground shaking during the 2011 M_w 5.8 Mineral, VA, earthquake given the earthquake's moderate size and epicentral distance (Hough, 2012; McNamara et al., 2014). The city is 130 km from the epicenter (Figure 1a), and the peak horizontal ground acceleration of 0.11 g on bedrock at nearby Reston, VA (Wells et al., 2015), would normally not be expected to cause significant damage. Nonetheless, the earthquake caused much-publicized damage to the Washington Monument (Nikolaou et al., 2011; Wells et al., 2015), the Washington National Cathedral (http://cathedral.org/earthquake/eq2/), the Sherman building at the Armed Forces Retirement Home (Swift, Daw, & Burgess, 2015, Swift, Daw, & Burke, 2015), the main building ("Castle") of the Smithsonian Institution (Clough, 2014), and other buildings (Horton et al., 2015). Minor damage in the city included rotated or toppled gravestones, cracked masonry and chimneys, and broken architectural elements such as facing stones. It was fortunate that no serious injuries or deaths occurred in the city from broken and falling masonry.

The 2011 earthquake damage in Washington, DC, highlights three factors that make cities in central and eastern North America vulnerable to moderately large earthquakes (M~5.5–6.5), which are a more common hazard than large earthquakes like the 1811–1812 M~7–7.5 Mississippi Valley earthquakes (e.g., Hough & Page, 2011; Johnston & Schweig, 1996; Nuttli, 1973; Tuttle et al., 2002) or the 1886 M~7 Charleston, SC, earthquake (Dura-Gomez & Talwani, 2009; Dutton, 1889; Wong et al., 2005). First, low seismic attenuation in central and eastern North America causes earthquakes to expose much wider areas to potentially damaging ground motions than in geologically young regions like much of western North America (e.g., Bockholt et al., 2015; Frankel et al., 1990; McNamara et al., 2014; Nuttli, 1973). Second, many older buildings there were constructed before modern seismic building codes were in effect.

Geophysical Research Letters



Figure 1. (a) Map of east-central U.S. showing Washington, DC (DC), and labeled geologic provinces (Bailey et al., 2016), including the Atlantic Coastal Plain (ACP) in yellow and green denoting Mesozoic rift basins. The blue focal mechanism is the M_w 5.8 Mineral, VA, earthquake of 2011, and the red dots are other earthquakes. Earthquakes west of Richmond, VA (Rd), form the Central Virginia seismic zone, from which three earthquakes were used in our analysis. R = Reston, VA. (b) Map of Washington, DC, showing Piedmont rocks (pink) and ACP strata and other unconsolidated deposits (yellow) (Darton, 1950; Southworth & Denenny, 2006; Powars, Catchings et al., 2015, Powars, Edwards et al., 2015). The red dots show seismometer sites, the black shapes show major buildings, and the rivers are gray. AFRH = Armed Forces Retirement Home (Sherman building); CA = Capitol; LM = Lincoln Memorial; NA = National Airport; NC = Washington National Cathedral; PG = Pentagon; WA = Washington Monument; WH = White House. The red line A-A' locates the A-A' cross section. (c) Cross section A-A' showing ACP strata and unconsolidated deposits increasing in thickness from near Rock Creek to about 250 m thick at the southeast edge of the city. A thin layer of unconsolidated deposits is also present near the Washington National Cathedral. (d) Model of the amplification effects due to thin layers. The two models assume layers with velocities of 400 m/s and thicknesses of 100 m and 50 m. Each layer produces a primary resonance peak (P100 and P50) and higher-frequency peaks at multiples of the primary frequency (e.g., H1, H2, and H3 for the 100 m thick layer) as predicted by equation (1).

A third factor, investigated here, is that shallow sediments and unconsolidated deposits often overlie highvelocity Mesozoic or older rocks in central and eastern North America, which creates large velocity and density contrasts that contribute to amplification of ground shaking (e.g., Baise et al., 2016; Field et al., 1990; Fischer et al., 1995; Motazedian et al., 2011; Yilar et al., 2017). This is the case in parts of Washington, DC, and in other central and eastern U.S. cities situated on Atlantic Coastal Plain (ACP) and Mississippi Embayment (ME) strata (e.g., Bodin & Horton, 1999; Jaumé & Ghanat, 2015). This amplification effect also applies to eastern U.S. and Canadian cities outside of the extent of the ACP and ME strata but situated on other low velocity sediments (e.g., Baise et al., 2016; Field et al., 1990; Fischer et al., 1995; Motazedian et al., 2011). This local amplification can raise widespread, moderate ground motions from modest-sized earthquakes to damaging levels. Amplified ground motions in Washington, DC, during the 2011 earthquake are suggested by the anomalously high intensities in "Did-You-Feel-It" reports (Hough, 2012) and the significant damage in the city. The 1988 M_w 5.9 Saguenay earthquake in southern Canada likewise caused damages in distant cities, with amplification by shallow soils contributing to damages in Quebec city about 150 km from the epicenter and in Montreal about 340 km from the epicenter (Mitchell et al., 1990).

Most of Washington, DC, is underlain by ACP and other unconsolidated deposits overlying metamorphic and igneous basement rocks (Fleming et al., 1994; Southworth & Denenny, 2006). ACP strata in the city begin near Rock Creek and gradually thicken to about 250 m at the southeast edge of the city (Figures 1b and 1c; Darton, 1950; Powars, Catchings et al., 2015; Powars, Edwards et al., 2015). Sedimentary deposits also underlie the area around the Washington National Cathedral (Figure 1b), and parts of the city are on man-made fill (Fleming et al., 1994; Southworth & Denenny, 2006; Wells et al., 2015).

Seismic wave amplitudes can increase substantially as they pass from relatively high-impedance (velocity \times density) basement rocks into low-impedance shallow layers (Borcherdt, 1970; Haskell, 1960; Shearer & Orcutt, 1987; Singh et al., 1988). Reverberations in shallow strata further increase amplitudes and

durations of ground shaking at specific resonant frequencies (F_n) defined by the thickness (h) and shear wave velocity (V_s) of the shallow strata:

$$F_n = (2n+1)V_s/4h \tag{1}$$

where *n* is the harmonic number (0 = primary peak; 1 = first harmonic, etc.) (e.g., Shearer & Orcutt, 1987; van der Baan, 2009). Thin surface layers can therefore cause a series of resonance peaks that decrease in amplitude at higher frequencies because of frequency-dependent attenuation (Figure 1d).

2. Measuring Amplification Effects

We deployed 27 seismometers throughout Washington, DC, between November 2014 and September 2015 to measure seismic wave amplifications caused by the shallow sedimentary deposits (Figure 1b and Table S1 in the supporting information). We computed spectral ratios of ground shaking relative to the average at four bedrock sites northwest of downtown, thus using the seismic waves reaching the bedrock sites as a proxy for the seismic waves entering the base of the shallow deposits. Identical 2 Hz sensors recorded at 100 samples/s produced usable signals from about 0.1 Hz (10 s period) to 25 Hz or more. Sensor response is reduced at low frequencies, but recordings with adequate signal-to-noise ratios are still useable. The response of the instruments was removed by dividing each site's Fourier frequency spectrum by the average spectrum at the reference sites. Instruments were located in open areas using battery power (e.g., graveyards) or at small buildings for power and security but away from large structures that would introduce significant building vibrations. Most sensors were buried in soil next to buildings, but a few sensors were placed inside small buildings on concrete floors with sand bags on top.

The relative lack of seismicity near Washington, DC, required us to rely on weak ground motions from distant earthquakes. We are not aware of reported liquefaction or pronounced ground settlement in Washington, DC, that indicates strong nonlinear effects in the city during the 2011 earthquake. However, small amounts of nonlinear soil response could reduce the peak amplifications from what we document using weak motions (e.g., Banab et al., 2012) and could slightly shift the frequency of the resonance peaks. Our results nonetheless identify the frequency range of amplification effects and highlight their importance, even if strong amplifications are partially mitigated by some nonlinearity in the soil response.

We analyzed recordings from 30 earthquakes having high signal-to-noise ratios (Figure 2 and Table S2), with low frequencies (0.1 to ~3 Hz) coming from teleseisms, middle frequencies (~0.5 to ~7 Hz) from regional earthquakes, and high frequencies (~1 to 25 Hz) from local earthquakes. We used 100 s windows for teleseisms and 50 or 25 s time windows for regional or local earthquakes (Figure 2). Some teleseisms with long durations allowed for analysis of several consecutive time windows. To provide adequate signal-to-noise ratios in this noisy urban area, we used time windows that spanned the strongest recorded ground motions from each event (e.g., Figure 2).

We applied a cosine taper to 10% of each end of the analysis windows, computed the Fourier spectrum of each component of ground motion, smoothed the spectra over 2.5 Hz, and then used the vector sum of the N and E spectra to produce a single horizontal spectrum. We removed frequency components with amplitudes less than twice that of a preearthquake noise window processed in an identical manner. The frequencies used from each earthquake at a site thus depended on the signal-to-noise ratio. Each data point on the spectra (Figures 3 and S1 in the supporting information) is the spectral ratio relative to the average of the four bedrock sites of a single frequency component during one earthquake, with the geometric mean of these points providing the summary amplification function at each site. The bedrock sites show spectral ratios near 1 between frequencies of 0.1 and about 5 Hz, above which each site shows some variation (Figure S1). We used the average signal from four reference sites to minimize the influence of a single site. We required two or more bedrock sites to be operating during any earthquake we analyzed, with few earthquakes having less than three reference sites operating.

3. Amplifications in Washington, DC

Inspection of the seismograms shows that sites on ACP strata had larger amplitudes and longer durations of ground shaking than bedrock sites in some frequency ranges (Figure 2). Nearly all sites on 15 m or more of ACP strata show large fundamental amplification peaks in the 0.7 to 4.0 Hz frequency range and little or



Figure 2. (a) Recordings of a M_d 3.0 earthquake beneath western North Carolina at two of our bedrock sites (DC01 and DC02) and two of our sites on moderately thick ACP strata (DC05 and DC29; ~61 and ~32 m, respectively). Note the larger amplitudes and longer durations of the recordings at sites on the ACP strata. The dashed lines show the 50 s analysis window used for this earthquake. Data have a 0.6 to 3 Hz band-pass filter for display. Amplitudes are raw data (counts) from the recording instrument. (b) Records from a M_w 6.8 earthquake from southern Alaska at two of our bedrock sites (DC02 and DC08) and at two sites on thick ACP strata (DC34 and DC36; ~169 m and ~150 m of ACP, respectively). The dashed lines show the 100 s analysis window encompassing the strongest shaking. Note the larger amplitudes and longer durations of the recordings made on ACP strata. Data have a 0.1 to 1 Hz band-pass filter for display.

no amplification at lower frequencies (Figure 3). Higher-frequency resonance peaks generally show decreasing amplifications with increasing frequency. Sites on thicker strata (>130 m) show a low-frequency, fundamental peak caused by resonance between the surface and the bedrock interface, and sometimes larger higher-frequency peaks presumably due to harmonics of the main peak and resonances from reflective layers within the ACP strata. Sites with less than ~15 m of ACP strata show flat responses similar to bedrock sites, with only minor amplification peaks at high frequencies caused by thin layers or

10.1002/2017GL075517

Geophysical Research Letters



Figure 3. Spectral ratios at sites with well-defined resonance peaks, arranged by increasing thickness of ACP strata and unconsolidated deposits (i.e., depth to bedrock). The site number and depth to bedrock are listed above each graph. Each black dot represents the spectral ratio of a single frequency during a single earthquake, the red dots are the geometric mean of the data points at each frequency component, and the green dots show one standard deviation from the mean. Strong resonance peaks occur at nearly all sites on ACP strata, and the frequency of the primary resonance peak decreases with increasing ACP thickness as expected from equation (1).

AGU Geophysical Research Letters



Figure 4. (a–f) Comparison of observed site responses over a range of ACP thicknesses (red lines) compared to modeled site responses from Tilashalski et al. (2015) (blue lines). Note that the models predict well the primary and first resonance peaks, suggesting that ACP strata are the primary cause of the amplification peaks. The best matches often are with modeled responses that have 10 to 25 m greater thickness of ACP strata because the thick layer of weathered bedrock (saprolite) causes bedrock velocities to be reached at depths below the lithologic boundary. A model with thicker ACP strata might fit the observed data better in Figure 4f, but 130 m was the thickest model for which there was geotechnical data.

vibrations of adjacent small buildings. The frequencies of the primary resonance peaks decrease with increasing thickness of ACP strata (Figure 3), as predicted by equation (1).

Comparisons with 1-D site response models indicate that ACP strata are responsible for most of the observed amplifications, without requiring 3-D effects such as surface waves or basin edge effects (e.g., Bard & Bouchon, 1985; Frankel et al., 1991, 2009; Kawase, 1996). Tilashalski et al. (2015) modeled amplification effects at 122 sites in Washington, DC, using depth to bedrock (Darton, 1950; Powars, Catchings et al., 2015; Powars, Edwards et al., 2015), measured shear wave velocities (Kayen et al., 2015), and unpublished geotechnical data. The geotechnical data were collected primarily near large buildings in busy parts of the city, whereas our instruments were in quieter areas away from large buildings. The modeled site responses are thus at different sites than our seismometers, and they showed some variation even for similar ACP thicknesses. The ACP thicknesses also have some uncertainty because they are based on extrapolation of relatively sparse drill hole data (Darton, 1950; Powars, Edwards et al., 2015). Nonetheless, models with similar ACP thicknesses as our sites match well the primary and next higher frequency resonance peaks and the flat responses at lower frequencies (Figures 4a–4f). Peaks between the first two major resonance peaks likely originate from reverberations from reflective layers within the ACP strata.

The best matches to observed site responses often are from models with ACP thicknesses 10 to 25 m greater than at the site (Figures 4a–4f), presumably due to thick saprolite (decomposed bedrock from chemical weathering) causing bedrock velocities to be reached at depths below the lithologic surface determined from drill holes. Saprolite in the Washington, DC, area has been observed with thicknesses up to 50 m on exposed bedrock (e.g., Froelich, 1975), and thicknesses of 15 m are common on bedrock beneath ACP sediments (e.g., Pavich & Obermeier, 1985).

Measured site responses above about 7 Hz generally show smaller spectral ratios than modeled responses (Figures 4a–4f), likely because scattering of high-frequency seismic waves by shallow structures and uneven interfaces is not accounted for in the 1-D site response models. This observation suggests that in areas with thick ACP strata the attenuation effect is greater than the amplification effect and the strata could reduce ground motions at high frequencies. However, this effect in the Fourier domain may not always indicate decreased response spectra, which are influenced by a broader frequency range of ground motions.

We computed an average shear wave velocity for the ACP strata beneath each site by using equation (1), the observed frequency of the main resonance peak, and the ACP thicknesses estimated from the drill hole data (but shifted to 10 m greater depth because of the saprolite). Using the average velocities for ACP strata of



Figure 5. (a) Graph of the frequency of the primary resonance peak versus thickness of ACP strata, with the predicted values from equation (1) and our derived velocity function shown as a black line. The blue dots are the observed data, the red dots are the observed data with 10 m added to the ACP thickness because of saprolite (see text), and the black dots are from the models based on geotechnical data (Tilashalski et al., 2015). The anomalously low-resonant frequency at site DC28 is likely because it is on man-made fill with an anomalously low velocity. (b) The velocity function for ACP strata was derived by matching the predicted resonant frequencies based on ACP thickness (black line in part Figure 5a) using equation (1).

different thicknesses, we computed a velocity function for the ACP strata that provided a good fit to the observations and models (Figures 5a and 5b). Polynomial equations, even to sixth order, did not fit the data well, so we used trial and error to modify the velocities at different depths to match the data. The resulting velocity function is 350 m/s in the upper 20 m, increases to 760 m/s at about 75 m depth, and reaches 1,000 m/s below about 120 m (Figure 4h). We emphasize that this velocity function is only loosely constrained and is nonunique, with trade-offs between the shallow and deeper velocities. Variations in ACP velocities across the city also make use of our velocity function problematic at any specific site. Nevertheless, the frequency of the primary resonance peak throughout the city can be estimated based on the thickness of the ACP strata (Figure 5a), or vice versa.

While amplification peaks reach factors of 10 or more at some sites, our results are derived using only weak motion data. Strong motion amplifications could be lowered by pervasively nonlinear response (e.g., Banab et al., 2012; Chin & Aki, 1991; Seed et al., 1976). While clear nonlinear effects such as liquefaction were not reported in Washington, DC, during the 2011 earthquake, sediments can experience significant nonlinear effects even at low to moderate accelerations (e.g., Banab et al., 2012; Chin & Aki, 1991; Rubinstein, 2011). "Did-You-Feel-It" intensity data suggest that local amplifications in the Washington, DC, area in 2011 were generally less than about 1.5 MMI units, corresponding to a peak acceleration amplification of about 3 (Hough, 2000). A nonlinear response of ACP strata in the region could explain why this factor was not as high as we measured from weak motions. Nonetheless, although pervasive nonlinearity could slightly shift and decrease the fundamental resonance peak, our results make it clear that amplification effects contribute significantly to hazard.

4. Discussion

4.1. Resonant Frequencies and Building Heights

A major factor in a building's resistance to earthquake damage is the relationship between the resonant frequencies of a building and the ground shaking, with a matching of frequencies increasing the chances of damage. An extreme example was the 1985 earthquake in Mexico City, where shallow sediments and trapping of surface waves in the underlying basin caused large amplifications of ground motions (summary in Flores-Estrella & Lomnitz, 2007). The matching of resonant frequencies during the earthquake was cited as contributing to severe damage to 5- to 15-story buildings while buildings of other heights sustained less damage (e.g., Flores et al., 1987; Seligman et al., 1989; Singh et al., 1988).

Our results suggest a matching of the resonant frequencies of buildings and amplified ground shaking in Washington, DC. Resonant frequencies of buildings are approximately inversely proportional to their height (Chopra & Goel, 2000; Dym & Williams, 2007; Ellis, 1980; Goel & Chopra, 1997, 1998). Two- to five-story houses and small commercial buildings are prevalent in Washington, DC, and height restrictions limit most

buildings in the downtown business district to heights less than about 40 m. Empirical studies of concrete shear-wall and steel-frame buildings suggest resonant frequencies in the 0.4 to 5 Hz range for the 6 to 40 m tall buildings that dominate the city (Chopra & Goel, 2000; Dym & Williams, 2007; Goel & Chopra, 1997, 1998). These likely resonant frequencies overlap with the frequency range of amplified ground motions documented here. This matching of frequencies suggests that the amplified ground motions likely contributed to building damages during the 2011 earthquake and to the anomalously high perceived shaking reflected in the USGS "Did-You-Feel-It" intensity reports (Hough, 2012). There were no building collapses in Washington, DC, during the 2011 earthquake, but even without building collapse, amplified shaking in future earthquakes or in other cities makes it more likely that poorly attached veneers or architectural elements will sustain damage or fail, as happened to some buildings in Washington, DC, during the 2011 earthquake.

The buildings in Washington, DC, with the most damage during the 2011 earthquake, however, were mostly larger, historic buildings that likely have fundamental frequencies lower than the resonance peaks documented here. The Washington Monument provides a singular but important case study. Wells et al. (2015) concluded after an engineering analysis that the 169 m tall structure has a fundamental resonance mode of about 0.3 Hz (2.8–3.2 s period) and a second-order (S-shape) resonance mode of about 1.11 Hz (0.9 s period). However, damage was prevalent at the pyramidion top of the monument, the components of which have resonant frequencies in the 2.5 to 5 Hz range (0.2 to 0.4 s periods). Wells et al. (2015) found that shallow deposits beneath the monument amplified ground motions in a frequency range of 2.5 to 5 Hz (0.2 to 0.4 s periods), which overlaps the resonant frequencies of the pyramidion. They concluded that resonance of the upper elements of the building (pyramidion) in these amplified ground motions likely caused the concentration of damage there. It thus appears that amplification by shallow deposits was a significant factor contributing to the damage to the Washington Monument in the 2011 earthquake.

Matching of frequencies of amplified seismic waves with the resonant frequencies of specific building components may hold true for other large buildings in the city. In particular, the National Cathedral had damage concentrated in the flying buttresses and the relatively small, unreinforced spires and towers in the upper parts of the building (http://cathedral.org/earthquake/eq2/). Amplified ground motions were also suggested as contributing to the damage primarily to chimneys and other small elements at the Smithsonian "Castle" (Clough, 2014). Both the cathedral and the Smithsonian "Castle" are on thin layers of sediment (e.g., Clough, 2014; Mark et al., 2001) consistent with high frequencies of ground shaking being amplified. We can only speculate, but the pattern of damage in these buildings may be similar to the Washington Monument, with damage to the smaller architectural elements being exacerbated by amplification of high-frequency ground shaking caused by thin layers of sediment.

In contrast to these buildings, the badly damaged Sherman Building at the Armed Forces Retirement Home may have been shaken by ground motions amplified near its fundamental resonant frequency. This was among the most badly damaged buildings in the Washington-Baltimore area during the 2011 earthquake (Swift, Daw, & Burgess, 2015, Swift, Daw, & Burke, 2015). The building is masonry with a ~37 m tall tower and a main building a little more than half that height. Given the heights of the building and tower, their resonant frequencies are likely in the 1 to 3 Hz range. The building sits on about 89 m of ACP strata above crystalline bedrock, and the seismometer we located 280 m southwest of the building on about 73 m of strata showed a strong amplification peak at a frequency of about 1.8 Hz (site DC06 in Figure S1). These numbers suggest that the building was shaken by ground motions that were amplified substantially at about the resonant frequency of the building.

4.2. Implications for Central and Eastern North America Earthquake Hazards

Damages in Washington, DC, during the Mineral earthquake and to eastern Canadian cities in the Saguenay earthquake demonstrate that moderate ground motions can be locally amplified to damaging levels, with implications for many cities in central and eastern North America. Our results show that ACP sediments and other unconsolidated deposits are the dominant factor in amplifying ground shaking in Washington, DC, and likely contributed to the damages there. Strong resonance effects (reverberations) are created by the large contrast in material properties between the unconsolidated shallow deposits and the underlying metamorphic or igneous basement rocks. Similar amplification effects have been documented in other central and eastern North American cities such as Boston, MA, New York City, Providence, RI, Charleston, SC, and

Ottawa, Canada (e.g., Baise et al., 2016; Banab et al., 2012; Braganza et al., 2017; Field et al., 1990; Fischer et al., 1995; Jaumé & Ghanat, 2015; Motazedian et al., 2011; Yilar et al., 2017), and presumably would be a factor in other central and eastern North American cities built entirely or partially on shallow ACP and other sedimentary deposits overlying crystalline rocks, such as Trenton, NJ, Wilmington, DE, Baltimore, MD, Richmond, VA, and Columbia, SC. Deposits similar to the ACP strata also underlie cities in the Mississippi Valley and Gulf Coast of the U.S., notably Memphis, TN (Bodin & Horton, 1999). Our sites had a maximum ACP thickness of about 200 m, but we can infer that thicker ACP strata will cause resonance peaks at lower frequencies, as well as high-frequency peaks caused by layering within the ACP strata (e.g., Jaumé & Ghanat, 2015; Pratt & Magnani, 2017).

Local amplification by unconsolidated strata is a special concern for cities located in stable continental interiors and passive margin settings. Low attenuation of seismic energy in these settings can extend similar levels of ground shaking to much greater distances than in geologically younger areas like parts of western North America (e.g., Frankel et al., 1990; Horton & Williams, 2012; McNamara et al., 2014; Nuttli, 1973). The 2011 Mineral and 1988 Saguenay earthquakes show that moderate ground motions can be locally amplified to levels that can cause substantial damage, even if the epicenter was in a rural area distant from major cities. A larger earthquake, or a moderate earthquake near a major metropolitan area, could produce much greater damage if local amplifications like we document in Washington, DC, substantially increase the strength of shaking. Furthermore, the increasing seismicity due to induced earthquakes like the M_w 5.8 Pawnee, OK, earthquake (e.g., Chen & Nakata, 2017) suggests that moderate ground shaking may be more common in the future. Our results in Washington, DC, thus highlight the importance of local amplification of ground motions leading to future earthquake damages in central and eastern North American cities and in passive margin and midplate regions in general.

Acknowledgments

We thank the property owners and managers who helped choose the seismometer sites and provided access. The instruments were from the IRIS PASSCAL instrument pool, and we thank Noel Barstow, Mouse Reusch, and Alissa Scire for preparing and shipping the seismometers and helping with the data processing and archiving. We also thank David Powars (USGS) for the ACP thickness estimates in Table S1 and Terrence Paret of Wiss, Janney, Elstner Associates, Inc. for discussions of building damage during the 2011 earthquake. The paper benefited from insightful reviews by Gail Atkinson, Eric Thompson, and an anonymous reviewer. This work was supported by the USGS Earthquake Hazards **Reduction Program Internal Funds and** External Grants Award G13AP00076. Data from the experiment, including the supporting information for this paper, are archived and publicly available at the IRIS Data Management Center (http://ds.iris.edu/ds/nodes/dmc/) with network code ZN (2014-2015) and nicknamed "DCShake." The station locations, list of earthquakes used in the analysis, and plots of the site responses are included in the supporting information for this article.

References

- Bailey, C. M., Sherwood, W. C., Eaton, L. S., & Powars, D. S. (Eds.) (2016). The Geology of Virginia. Virginia Museum of Natural History, Special Publication (Vol. 18, 538 pp.).
- Baise, L. G., Kaklamanos, J., Berry, B., & Thompson, E. M. (2016). Soil amplification with a strong impedance contrast: Boston, Massachusetts. *Engineering Geology*, 202, 1–13. https://doi.org/10.1016/j.enggeo.2015.12.016
- Banab, K. K., Kolaj, M., Motazedian, D., Sivathayalan, S., Hunter, J. A., Crow, H. L., ... Pyne, M. (2012). Seismic site response analysis for Ottawa, Canada: A comprehensive study using measurement and numerical simulations. *Bulletin of the Seismological Society of America*, 102, 1976–1993. https://doi.org/10.1785/0120110248
- Bard, P. Y., & Bouchon, M. (1985). The two-dimensional resonance of sediment-filled valleys. Bulletin of the Seismological Society of America, 75, 519–541.
- Bockholt, B. M., Langston, C. A., & Withers, M. (2015). Local magnitude and anomalous amplitude distance decay in the Eastern Tennessee Seismic Zone. Seismological Research Letters, 86, 1040–1050.

Bodin, P., & Horton, S. (1999). Broadband microtremor observation of basin resonance in the Mississippi embayment, central US. *Geophysical Research Letters*, 26(7), 903–906. https://doi.org/10.1029/1999GL900146

- Borcherdt, R. D. (1970). Effects of local geology on ground motion near San Francisco Bay. Bulletin of the Seismological Society of America, 60, 29–61.
- Braganza, S., Atkinson, G. M., & Molnar, S. (2017). Assessment of the spatial variability of site response in southern Ontario. Seismological Research Letters, 88(5), 1415–1426. https://doi.org/10.1785/0220170042
- Chen, X., & Nakata, N. (2017). Preface to the focus section on the 3 September 2016 Pawnee, Oklahoma, earthquake. Seismological Research Letters, 88(4), 953–955. https://doi.org/10.1785/0220170078
- Chin, B.-H., & Aki, K. (1991). Simultaneous study of the source, path, and site effects on strong ground motion during the 1989 Loma Prieta earthquake: A preliminary result on pervasive nonlinear site effects. *Bulletin of the Seismological Society of America*, *81*, 1859–1884.
- Chopra, A. K., & Goel, R. K. (2000). Building period formulas for estimating seismic displacements. Earthquake Spectra, 16(2), 533–536. https://doi.org/10.1193/1.1586125
- Clough, G. W. (2014). Back to back: The earthquake and hurricane of 2011. In A. E. Charola, C. Wegener, & R. J. Koestler (Eds.), Unexpected— Earthquake 2011 Lessons to be Learned, Smithsonian Contributions to Museum Conservation, Number 4 (pp. 1–18). Washington, DC: Smithsonian Institution Scholarly Press. Retrieved from http://opensi.si.edu/index.php/smithsonian/catalog/book/47
- Darton, N. H. (1950). Bedrock surface of the District of Columbia and vicinity. U. S. Geological Survey Professional Paper 217 (42 pp.). Dura-Gomez, I., & Talwani, P. (2009). Finding faults in the Charleston area, South Carolina: 1. Seismological data. Seismological Research Letters. 80, 883–900. https://doi.org/10.1785/gssrl.80.5.883
- Dutton, C. E. (1889). The Charleston earthquake of August 31, 1886. In Ninth Annual Report of the U.S. Geological Survey, (pp. 203–528). Dym, C. L., & Williams, H. E. (2007). Estimating fundamental frequencies of tall buildings. *Journal of Structural Engineering*, 133, 1479–1483. https://doi.org/10.1061/(ASCE)0733-9445(2007)133:10
- Ellis, B. R. (1980). An assessment of the accuracy of predicting the fundamental natural frequencies of buildings and the implications concerning the dynamic analysis of structures. *Proceedings of the Institution of Civil Engineers, Part 2, 69,* 763–776.
- Field, E. H., Hough, S. E., & Jacob, K. H. (1990). Using microtremors to assess potential earthquake site response: A case study in Flushing Meadows, New York City. *Bulletin of the Seismological Society of America*, 80, 1456–1480.
- Fischer, K. M., Salvati, L., Hough, S. E., Gonzalez, E., Nelson, C. E., & Roth, E. G. (1995). Sediment-induced amplification in the northeastern United States: A case study in Providence, RI. *Bulletin of the Seismological Society of America*, 85, 1388–1397.

Fleming, A. H., Drake, A. A., Jr., & McCartan, L. (1994). Geologic map of the Washington West Quadrangle, District of Columbia, Montgomery and Prince Georges Counties, Maryland, and Arlington and Fairfax Counties, Virginia. U.S. Geological Survey Geologic Quadrangle Map GQ-1748, scale 1:24,000, 1 sheet.

Flores, J., NoVaro, O., & Seligman, T. H. (1987). Possible resonance effect in the distribution of earthquake damage in Mexico City. Nature, 326(6115), 783–785. https://doi.org/10.1038/326783a0

Flores-Estrella, H., & Lomnitz, C. (2007). Seismic response of the Mexico City basin: A review of 20 years of research. Natural Hazards, 40, 357–372. https://doi.org/10.1007/s11069-006-0034-6

Frankel, A., Hough, S., Friberg, P., & Busby, R. (1991). Observations of Loma Prieta aftershocks from a dense array in Sunnyvale, California. Bulletin of the Seismological Society of America, 81, 1900–1922.

Frankel, A., McGarr, A., Bicknell, J., Mori, J., Seeber, L., & Cranswick, E. (1990). Attenuation of high-frequency shear waves in the crust: Measurements from New York state, South Africa, and southern California. *Journal of Geophysical Research*, 95(B11), 17,441–17,457. https://doi.org/10.1029/JB095iB11p17441

Frankel, A., Stephenson, W., & Carver, D. (2009). Sedimentary basin effects in Seattle, Washington: Ground-motion observations and 3D simulations. Bulletin of the Seismological Society of America, 99, 1579–1611. https://doi.org/10.1785/0120080203

Froelich, A. J. (1975). Map showing thickness of overburden, District of Columbia. U.S. Geological Survey Open-File Report 75–538, scale 1:24,000. Retrieved from https://pubs.er.usgs.gov/publication/ofr75538%3e

Goel, R. K., & Chopra, A. K. (1997). Period formulas for moment resisting frame buildings. *Journal of Structural Engineering*, 123(11), 1454–1461. https://doi.org/10.1061/(ASCE)0733-9445(1997)123:11

Goel, R. K., & Chopra, A. K. (1998). Period formulas for concrete shear wall buildings. *Journal of Structural Engineering*, 124(4), 426–433. https://doi.org/10.1061/(ASCE)0733-9445(1998)124:4

Haskell, N. A. (1960). Crustal reflection of plane SH waves. Journal of Geophysical Research, 65(12), 4147–4150. https://doi.org/10.1029/ JZ065i012p04147

- Horton, J. W. Jr., Chapman, M. C., & Green, R. A. (2015). The 2011 Mineral, Virginia, earthquake, and its significance for seismic hazards in eastern North America—Overview and synthesis. In J. W. Horton, Jr., M. C. Chapman, & R. A. Green (Eds.), The 2011 Mineral, Virginia, Earthquake and its Significance for Seismic Hazards in Eastern North America, Geological Society of America Special Paper (Vol. 509, pp. 1–25). https://doi.org/10.1130/2015.2509
- Horton, J. W. Jr., & Williams, R. A. (2012). The 2011 Virginia earthquake: What are scientists learning? *Eos, Transactions, American Geophysical Union*, *93*, 317–318. https://doi.org/10.1029/2012EO330001

Hough, S. E. (2000). On the scientific value of "unscientific" data. Seismological Research Letters, 71, 483–485.

Hough, S. E. (2012). Initial assessment of the intensity distribution of the 2011 M_w5.8 Mineral, Virginia, earthquake. Seismological Research Letters, 83, 649–657. https://doi.org/10.1785/0220110140

Hough, S. E., & Page, M. (2011). Towards a consistent model for strain rate accrual and release for the New Madrid Seismic Zone. Journal of Geophysical Research, 116, B03311. https://doi.org/10.1029/2010JB007783

Jaumé, S. C. & Ghanat, S. T. (2015). Translating observed weak motion site response into predicted strong motion in Charleston, South Carolina, USA. 6th International Conference on Earthquake Geotechnical Engineering (8 pp).

Johnston, A. C., & Schweig, E. S. (1996). The enigma of the New Madrid earthquakes of 1811–1812. Annual Review of Earth and Planetary Sciences, 24(1), 339–384. https://doi.org/10.1146/annurev.earth.24.1.339

Kawase, H. (1996). The cause of the damage belt in Kobe: "The basin-edge effect," constructive interference of the direct S-wave with the basin-induced diffracted/Rayleigh waves. Seismological Research Letters, 67(5), 25–34. https://doi.org/10.1785/gssrl.67.5.25

Kayen, R. E., Carkin, B. A., Corbett, S. C., Zangwill, A., Estevez, I., & Lai, L. (2015). Shear wave velocity and site amplification factors for 25 strong-motion instrument stations affected by the M5.8 Mineral, Virginia, earthquake of August 23, 2011. USGS Open-File Report 2015-1099 (66 pp).

Mark, R., Richards, R., Jr., & Mark, C. (2001). Geomechanics of large stone structures: A case history from the Washington National Cathedral. Proceedings of the 38th U.S. Rock Mechanics Symposium, D.C. Rocks 2001, Washington, DC: 7–10 July, 2001. Retrieved from http://www. cdc.gov/niosh/mining/userfiles/works/pdfs/golss.pdf

McNamara, D. E., Gee, L., Benz, H. M., & Chapman, M. (2014). Frequency-dependent seismic attenuation in the eastern United States as observed from the 2011 central Virginia earthquake and aftershock sequence. *Bulletin of the Seismological Society of America*, 104(1), 55–72. https://doi.org/10.1785/0120130045

Mitchell, D., Tinawi, R., & Law, T. (1990). Damage caused by the November 25, 1988, Saguenay earthquake. Canadian Journal of Civil Engineering, 17(3), 338–365. https://doi.org/10.1139/190-041

Motazedian, D., Hunter, J. A., Pugin, A., & Crow, H. (2011). Development of a Vs30 (NEHRP) map for the city of Ottawa, Ontario, Canada. *Canadian Geotechnical Journal*, 48(3), 458–472. https://doi.org/10.1139/T10-081

Nikolaou, S., Beyzaei, C. Z., Christie, D. W., & Lacy, H. S. (2011). The Washington Monument: Geology, Foundations, and Earthquake History, Special EERI Publication on the 2011 Virginia Earthquake. New York City: Mueser Rutledge Consulting Engineers. Retrieved from http:// www.eqclearinghouse.org/2011-08-23-virginia/2011/12/16/the-washington-monument-geology-foundations-earthquake-history/

Nuttli, O. W. (1973). Seismic wave attenuation and magnitude relations for eastern North America. Journal of Geophysical Research, 78(5), 876–885. https://doi.org/10.1029/JB078i005p00876

Pavich, M. J., & Obermeier, S. F. (1985). Saprolite formation beneath Coastal Plain sediments near Washington, DC. Geological Society of America Bulletin, 96(7), 886–900. https://doi.org/10.1130/0016-7606(1985)96%3C886:SFBCPS%3E2.0.CO;2

Powars, D. S., Catchings, R. D., Horton, J. W. Jr., Schindler, J. S., & Pavich, M. J. (2015). Stafford fault system: 120 million year fault movement history of northern Virginia. In J. W. Horton, Jr., M. C. Chapman, & R. A. Green (Eds.), *The 2011 Mineral, Virginia, Earthquake, and its Significance for Seismic Hazards in Eastern North America, Geological Society of America Special Paper* (Vol. 509, pp. 407–431). https://doi. org/10.1130/2015.2509

Powars, D. S., Edwards, L. E., Kidwell, S. M., & Schindler, J. S. (2015). Cenozoic stratigraphy and structure of the Chesapeake Bay region. In D. K. Brezinski, J. P. Halka, & R. A. Ortt, Jr. (Eds.), *Tripping From the Fall Line: Field Excursions for the GSA Annual Meeting, Baltimore, 2015, Geological Society of America Field Guide* (Vol. 40, pp. 171–229). https://doi.org/10.1130/2015.0040

Pratt, T. L., & Magnani, M. B. (2017). The effect of a sedimentary wedge on earthquake ground motions: The influence of eastern U.S. Atlantic Coastal Plain strata. Seismological Research Letters, 87, 647.

Rubinstein, J. L. (2011). Nonlinear site response in medium magnitude earthquakes near Parkfield, California. Bulletin of the Seismological Society of America, 101, 275–286. https://doi.org/10.1785/0120090396

Seed, H. B., Murarka, R., Lysmer, J., & Idriss, I. M. (1976). Relationships of maximum acceleration, maximum velocity, distance from source, and local site conditions for moderately strong earthquakes. *Bulletin of the Seismological Society of America*, 66, 1323–1342.

- Seligman, T. H., Alvarez-Tostado, J. M., Mateos, J. L., Flores, J., & NoVaro, O. (1989). Resonant response models for the Valley of Mexico—I; the elastic inclusion approach. *Geophysical Journal International*, *99*(3), 789–799. https://doi.org/10.1111/j.1365-246X.1989.tb02058.x
- Shearer, P. M., & Orcutt, J. A. (1987). Surface and near-surface effects on seismic waves: Theory and borehole seismometer results. Bulletin of the Seismological Society of America, 77, 1168–1196.
- Singh, S. K., Mena, E., & Castro, R. (1988). Some aspects of source characteristics of the 19 September 1985 Michoacan earthquake and ground motion amplification in and near Mexico City from strong motion data. *Bulletin of the Seismological Society of America*, 78, 451–477.
- Southworth, S., & Denenny, D. (2006). Geologic map of the national parks in the National Capital Region, Washington, DC: Virginia, Maryland, and West Virginia. U.S. Geological Survey Open-File Report 2005–1331. Retrieved from http://pubs.usgs.gov/of/2005/1331/
 - Swift, C., Daw, M., & Burke, L. (2015). Evaluations, repairs, and retrofit of the historic Sherman building in Washington, DC: following the 2011 Mineral, VA, earthquake. *Improving the Seismic Performance of Existing Buildings and Other Structures*, 2015, 97–108. https://doi.org/ 10.1061/9780784479728.009
- Swift, C. D., Daw, M. J., & Burgess, L. M. (2015). Evaluations, repairs and retrofits of the historic Sherman building in Washington, DC: following the 2011 Mineral, VA earthquake. Retrieved from http://keasthood.com/pdf/2015_ATC_SEI_Sherman_Paper_-_full_images.pdf or http:// cenews.com/article/9758/landmark-restoration-model
- Tilashalski, M., Olgun, C. G., Rodriguez-Marek, A., Godfrey, E., Chapman, M. C., Shamsalsadati, S., & Eddy, M. A. (2015). Regional geology and seismic site amplification in the Washington, DC, metropolitan area. *IFCEE*, 2015, 1278–1287. https://doi.org/10.1061/9780784479087.115
- Tuttle, M. P., Schweig, E. S., Sims, J. D., Lafferty, R. H., Wolf, L. W., & Haynes, M. L. (2002). Earthquake potential of the New Madrid seismic zone. Bulletin of the Seismological Society of America, 92, 2080–2089.
- van der Baan, M. (2009). The origin of SH-wave resonance frequencies in sedimentary layers. *Geophysical Journal International*, 178(3), 1587–1596. https://doi.org/10.1111/j.1365-246X.2009.04245.x
- Wells, D., Egan, J. A., Murphy, D. G., & Paret, T. (2015). Ground shaking and structural response of the Washington Monument during the 2011 Mineral, Virginia, earthquake. In J. W. Horton, Jr., M. C. Chapman, & R. A. Green (Eds.), The 2011 Mineral, Virginia, Earthquake, and its Significance for Seismic Hazards in Eastern North America, Geological Society of America Special Paper (Vol. 509, pp. 199–233). https://doi. org/10.1130/2015.2509
- Wong, I., Bouabid, J., Graf, W., Huyck, C., Porush, A., Silva, W., ... Knight, J. (2005). Potential losses in a repeat of the 1886 Charleston, South Carolina, earthquake. *Earthquake Spectra*, 21, 1157–1184. https://doi.org/10.1193/1.2083907
- Yilar, E., Baise, L. G., & Ebel, J. E. (2017). Using H/V measurements to determine depth to bedrock and Vs30 in Boston, Massachusetts. Engineering Geology, 217, 12–22. https://doi.org/10.1016/j.enggeo.2016.12.002